DISCUSSION PAPER SERIES

DP16064 (v. 2)

Managerial Talent and Managerial Practices: Are They Complements?

Audinga Baltrunaite, Giulia Bovini and Sauro Mocetti

ORGANIZATIONAL ECONOMICS



Managerial Talent and Managerial Practices: Are They Complements?

Audinga Baltrunaite, Giulia Bovini and Sauro Mocetti

Discussion Paper DP16064 First Published 22 April 2021 This Revision 31 May 2021

Centre for Economic Policy Research 33 Great Sutton Street, London EC1V 0DX, UK Tel: +44 (0)20 7183 8801 www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programmes:

• Organizational Economics

Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as an educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Copyright: Audinga Baltrunaite, Giulia Bovini and Sauro Mocetti

Managerial Talent and Managerial Practices: Are They Complements?

Abstract

We examine the role of managerial talent and its interaction with managerial practices in determining firm performance. We build a matched firm-director panel dataset for the universe of limited liability companies in Italy, tracking individuals across different firms over time. We define managerial talent as management's capacity to boost firms' total factor productivity, estimated using a two-way fixed effects model. Combining the data with survey information on a representative sample of firms, we then document that our measure of talent correlates with exante and ex-post indicators of ability, i.e. managers' educational attainment and their forecast precision with respect to the firm's future performance. Most important, we leverage information on the adoption of managerial practices within the firm to examine potential synergies between managerial talent and structured managerial practices, thus bridging two separate strands of the literature. While talent and structured practices do boost firm productivity on their own, there is evidence of complementarities between the two. These findings hold both in a cross-sectional setting and in a panel analysis that accounts for time-invariant firm heterogeneity. Overall, our results indicate that the effectiveness of managerial practices depends on managers' ability to implement them.

JEL Classification: G34, M10, D24

Keywords: Board Of Directors, Managers, corporate governance, productivity, managerial practices

Audinga Baltrunaite - audinga.baltrunaite@bancaditalia.it Bank of Italy and CEPR

Giulia Bovini - giulia.bovini@bancaditalia.it Bank of Italy

Sauro Mocetti - sauro.mocetti@bancaditalia.it Bank of Italy

Managerial Talent and Managerial Practices: Are They Complements?

Audinga Baltrunaite

Giulia Bovini

Sauro Mocetti

Bank of Italy^{*}

May 2021

Abstract

We examine the role of managerial talent and its interaction with managerial practices in determining firm performance. We build a matched firm-director panel dataset for the universe of limited liability companies in Italy, tracking individuals across different firms over time. We define managerial talent as management's capacity to boost firms' total factor productivity, estimated using a two-way fixed effects model. Combining the data with survey information on a representative sample of firms, we then document that our measure of talent correlates with ex-ante and ex-post indicators of ability, i.e. managers' educational attainment and their forecast precision with respect to the firm's future performance. Most important, we leverage information on the adoption of managerial practices within the firm to examine potential synergies between managerial talent and structured managerial practices, thus bridging two separate strands of the literature. While talent and structured practices do boost firm productivity on their own, there is evidence of complementarities between the two. These findings hold both in a cross-sectional setting and in a panel analysis that accounts for time-invariant firm heterogeneity. Overall, our results indicate that the effectiveness of managerial practices depends on managers' ability to implement them.

Keywords: Board of directors, managers, corporate governance, productivity, managerial practices.

JEL classification: G34, M10, D24.

^{*}We thank Matteo Bugamelli, David Card, Federico Cingano, Jan Eeckhout, Mike Elsby, Alessandra Fenizia, Silvia Giacomelli, Maria Guadalupe, Lawrence Katz, Patrick Kline, Enrico Moretti, Paolo Sestito, Eliana Viviano and participants at the Bank of Italy seminar (Rome, April 2019), Petralia Applied Economics Workshop (Petralia, Juy 2019), 34th Annual Congress of the European Economic Association (Manchester, August 2019), 34th Italian Conference of Labour Economics (Novara, September 2019), seminar at UC Berkeley (Berkeley, October 2019), "Recent trends in firm organization and firm dynamics" conference (Rome, December 2019) and seminar at the University of Turin (Turin, February 2020). The views and opinions expressed in this paper pertain to the authors only and do not represent in any way those of the Bank of Italy.

1 Introduction

The productivity differences among firms are strikingly large (Melitz, 2003; Syverson, 2011).¹ Several studies argue that such heterogeneity can be determined by management quality, both in terms of people who run the firm (e.g., Bertrand and Schoar, 2003) and practices adopted within the firm (e.g., Bloom and Van Reenen, 2010). The two strands of literature, however, have run on parallel paths: up to date, very little is known about the interaction between the talent of firm leaders and the adoption of structured managerial practices in influencing firm performance (Syverson, 2011; Gibbons and Henderson, 2013). This stands at a sharp contrast with the common wisdom that the effectiveness of tools may to a large extent depend on the ability of their masters. For example, while some practices in principal may appear optimal (e.g. promotion of competent employees), their efficiency in practice may largely depend on the talent of managers who adopt them (e.g. decide whether an employee is competent or not).

Following this rationale, we use unique administrative and survey data on Italian firms and their managers to explore the presence of complementarities between management talent and managerial practices in explaining the firm-level total factor productivity (TFP henceforth). First, we derive a measure of time-invariant and portable component of talent for individuals responsible for the key corporate decisions. To do so, we build a matched firmdirector dataset covering the universe of the non-micro limited liability companies in Italy and identities (and personal characteristics) of their directors. The data cover the period 2005-2018, tracking individuals across different firms over time.² Our proxy for managerial talent is based on the director fixed effect in a two-way fixed effects model – inspired by the work of Abowd et al. (1999) – and represents the individual contribution to the variation of the firms' TFP, conditional on firm fixed effects and sectoral and geographical

¹In a typical four-digit manufacturing industry in the U.S., establishments at the 90th percentile of total factor productivity distribution make almost twice as much output with the same inputs as plants at the 10th percentile (Syverson, 2004).

 $^{^{2}}$ We exclude firms constantly below the 20-employee-threshold in the period analyzed in order to consider companies above a minimum threshold of size and, therefore, complexity. Moreover, in this way we impose a common size threshold across the different data sources used in the paper.

non-parametric time trends. The average of the estimated director fixed effects at the firm level is, therefore, our measure of its management talent.

We find evidence that managerial talent matters for the firm productivity which rises when a better manager takes charge. The estimated impact is sizable: a one standard deviation change in management talent leads to about three fourths of a standard deviation variation in the firm performance. Including management talent in a regression of firms' TFP on firm-fixed effects increases the predictive power of the model by 19% (adjusted Rsquared rises from 0.54 to 0.64). This impact is in the ballpark of the estimates obtained by previous studies (Bertrand and Schoar, 2003; Graham et al., 2012).³ We also document the heterogeneity of the talent role across different firms and across different contexts in which they operate. Namely, the effect grows with firm size and age (and, arguably, the complexity of organizational processes to be managed within the firm) and it is larger in firms exposed to higher competition (e.g., in more agglomerated areas or in sectors more exposed to international trade).⁴

We then combine our data with information from a yearly survey conducted by the Bank of Italy. We document that our talent measure correlates with other (observable) proxies for leaders' talent: individuals we identify as talented managers more often possess adequate education or training, and are better in precisely forecasting the firm's performance.⁵ In other words, both ex-ante and ex-post ability measures validate the informational content in our talent indicator.

Finally, we use the information on management practices to explore their role – on their own or in combination with managerial talent – in determining firm performance. Namely, we

³The estimation of managerial talent via a high-dimensional two-way fixed effect model relies on the assumption that sorting of directors into companies is as good as random, conditional on firm time-invariant characteristics and other observed covariates. The empirical tests of this assumption lead to a conclusion that endogenous sorting based on the idiosyncratic value of the match or on the transitory component of firms' productivity does not seem to be relevant in this setting. Yet, there is some evidence of sorting based on the trend component of productivity, thus suggesting that our estimates of the contribution of managerial talent to a firm TFP could be slightly overstated in levels.

⁴These results mirror those by Bloom and Van Reenen (2010) with reference to the managerial practices. ⁵Malmendier and Tate (2005) and Ben-David et al. (2013) show that managerial overconfidence and miscalibration can account for significant corporate investment distortions.

exploit ad-hoc questions on firm organization and the adoption of managerial practices in the 2010 and 2019 waves of the survey. We find that the use of structured managerial practices boosts firm productivity on its own and, most importantly, there is evidence of synergies with managerial talent. Our results are sizable and statistically significant. The main impact comes from the use of specific schemes to measure performance, the proper definition of operational targets and the adoption of performance-based workforce promotion systems. Furthermore, the latter feature of structured management is observed both in the 2010 and 2019 survey waves, allowing us to estimate panel specifications that exploit within-firm variation and control for observed and unobserved firm heterogeneity. In quantitative terms, the adoption of a performance-pay system is associated to a nearly 9% increase in the TFP, while moving from the 25th to the 75th percentile of the managerial talent distribution is associated to a 11% increase of the TFP. Interestingly, the same increase of managerial talent leads to a further 8% increase of the TFP if associated to the adoption of performance-related workforce remuneration, indicating the presence of significant complementarities between management talent and performance-based pay.

Our paper is related to a growing literature examining the role of managers. In a seminal paper Bertrand and Schoar (2003) examine top executives (e.g., CEOs, CFOs, Presidents, etc.) who manage at least two firms in their sample period and find that the individual manager fixed effects are significantly correlated with firms' performance.⁶ Other studies exploit a similar strategy to identify manager fixed effects within a single firm or organization (Lazear et al., 2015; Fenizia, 2020).⁷

 $^{^{6}}$ In a similar vein, Lieberman et al. (1990) find that manager fixed effects are significant in explaining productivity variation in the U.S. and Japanese automobile industry. Interestingly, Graham et al. (2012) identify manager fixed effects using both individual-level regressions – where the dependent variable is manager compensation – and firm-level regressions – where the outcomes are different indicators of firm performance as in Bertrand and Schoar (2003). They find that manager fixed effects in compensation are significantly correlated with manager fixed effects estimated in the regression of firm outcomes.

⁷Instead of focusing on manager fixed effects, other related studies investigate the role of managers' characteristics on firm performance: Bennedsen et al. (2007) show that family CEOs have a negative causal effect on firm performance; Kaplan et al. (2012) document how differences in CEOs psychological traits explain the performance of the firms they manage; and Bandiera et al. (2020) build an individual-level index of behavior by parsing CEOs diaries and find that "leaders" are more likely to manage more productive and profitable firms. Partly related, Adams et al. (2018) and Bernile et al. (2018) examine the role of skill

Another strand of literature has emphasized the role of managerial practices. In a seminal paper, Ichniowski et al. (1997) find that the adoption of advanced management practices (e.g. incentive pay and employee participation in problem-solving teams) are significantly correlated with plant-level productivity. The interest in this topic has increased enormously thanks to the surveys managed by Bloom and Van Reenen and their research team, who collect information on managerial practices at the plant level for a wide set of industries and countries. Bloom and Van Reenen (2007) and Bloom and Van Reenen (2010) contain a comprehensive analysis of the relationship between management practices and productivity. Bloom et al. (2013) find a large causal role for such management practices in a field experiment with Indian textile plants.⁸

Our paper contributes to the existing literature along two main directions. First, our measure of managerial talent is close to Bertrand and Schoar (2003) and to the subsequent papers based on a similar two-way fixed effect model. We extend their analysis by exploiting additional sources of variation due to board-interlocking and, more importantly, by using a larger and more representative sample of firms. In fact, the empirical evidence on the importance of managers for firm performance is typically carried out on small and not representative samples: Bertrand and Schoar (2003) base their analysis on the sample of 500 CEOs of large publicly traded companies; Lazear et al. (2015) and Fenizia (2020) focus instead on a single large firm operating in service or public sector, respectively, with the sample size limited to several hundred individuals.⁹ The larger sample size allows us to explore the role of managers' quality in smaller firms and to examine heterogeneity of the effects. Last, but not least, we further enrich our data with survey information to examine correlation with other firm characteristics often unobserved by researchers, such as education of the

composition and diversity of the boards on firm performance.

⁸See Giorcelli (2019) for long-run causal effect of management on firm performance. See Lazear and Shaw (2007) for an analysis on human resource management practices like compensation, hiring practices, and teamwork.

⁹Similarly, the literature on managerial practices relies on detailed data on a moderate number of firms. For example, Bloom and Van Reenen (2007) explore managerial practices of around 700 firms in the U.S., the U.K., France, and Germany, while Bloom and Van Reenen (2010) - of around 6000 large firms in 17 countries worldwide.

firm leader or the firms' ability to forecast its future performance.

Second, we examine the complementarities between managerial talent and the use of managerial practices bridging the gap between two different strands of literature that have examined these factors separately.¹⁰ Previous research shows that the same "best practice" or innovation may produce heterogeneous results across firms, possibly due to the presence of complementarities among production inputs and/or different organizational features (Brynjolfsson and Milgrom, 2013). Our results show that talented managers – the actors responsible for main strategic and organizational choices within the firm – are able to extract more value added from the same managerial practice. Overall, our results indicate that able leaders are valuable to the firm not only because of wise decisions they make, but also because of how such decisions are put in practice.

The structure of the paper is as follows. Section 2 describes the data and variables. Section 3 discusses the construction of our measure of managerial talent and provides descriptive evidence. Section 4 examines complementarities between managerial talent and managerial practices. Section 5 concludes.

2 Data and variables

The analysis relies on three main datasets. First, we use the *Infocamere* database which is based on administrative data on the Italian firms gathered by provincial Chambers of Commerce. It contains information on the registration data of the universe of Italian limited liability firms. Most important, this dataset includes personal information on firms' stockholders and directors, i.e., name, surname and personal identification code. We use this information to identify the governance structure and the age, gender and place of birth of the directors.

Limited liability firms are typically organized under the traditional corporate governance

¹⁰One notable exception is Bender et al. (2018) who examine complementarities between managerial practices and managers, although the latter are not directly observed and proxied with the employees at the top quartile of the skill distribution. See Cornwell et al. (2021) for an analysis on the relationships between management practices and workforce composition.

model, whereby firm's stockholders appoint the governance body.¹¹ In particular, key corporate decisions may be concentrated in the hands of a single director (*amministratore unico*) or granted to a collegial governance body, i.e., the board of directors (*consiglio di amministrazione*). Director selection is essentially unregulated, as there are no restrictions regarding directors' independence, their stock ownership, previous experience or education.¹² While directors have the standard – advising and monitoring – duties, their *de facto* primary role is that of running the company as they are naturally small. Motivated by these considerations, we interchangeably use the terms *directors* and *managers* throughout the paper.

Second, we use the database managed by the *Cerved Group* which gathers balance sheet information of the universe of the Italian limited liability firms in the private non-financial sector. The dataset includes the value added and the revenues of the firm, its productive inputs and other anagraphic information such as firm's age, sector of economic activity and the municipality where the headquarter is located.¹³ We use balance sheet information to compute the TFP using the Levinsohn and Petrin (2003) estimator with the Ackerberg et al. (2015) correction.

Our main sample (i.e., the matched firm-director panel dataset) comprises all the firms included in the intersection of the *Infocamere* and *Cerved* databases for the years from 2005 to 2018 (the longest panel for both datasets) for which there are available data to compute measures of firm performance. We exclude from the analysis the micro firms and we restrict the analysis to the private non-financial sector.

Third, to validate our measure of managerial talent and to examine its complementarity with structured managerial practices, we merge our data with the Bank of Italy Survey of

¹¹The 2003 Italian corporate law reform introduced the possibility for joint-stock companies (*società per azioni*) to adopt a two-tiered model or a one-tiered model in alternative to the traditional one. The adoption of these new governance structures, however, remained remarkably low (less than 1% of firms in the universe of limited liability companies).

 $^{^{12}}$ See Baltrunaite et al., 2019 for descriptive analysis on the characteristics of directors in Italian companies.

¹³For obtaining information about firms' employees, we have also used firm-level data drawn from social security records (*INPS* dataset), which contains the distinction between (middle) managers, white- and blue-collar workers and their corresponding monthly wages. This enables us to construct efficiency units as alternative to total employment at the firm level, giving different weights to (middle) managers, white- and blue-collar, with weights being proportional to the relative wage premia.

Industrial and Service Firms (*Invind*), containing yearly information on firm organization for a representative sample of firms with at least 20 employees operating in the private nonfinancial sector. In particular, each wave contains firm's forecast about future performance, while the 2006 wave includes information on the education level of the firm leader. Concerning firm organization and managerial practices, the 2010 wave contains questions on the presence and the extent of team work in the production process, the adoption of performancerelated pay, and employees' involvement in the decision-making within the firm. Moreover, the 2019 wave contains a special section on the structured managerial practices, including questions drawn from the Management and Organizational Practices Survey (MOPS) described in Bloom et al. (2019). Specifically, the survey contains 8 management questions in three main areas: monitoring, targets, and incentives. The monitoring section asked firms about their collection and use of information to monitor and improve the production process. For example, the survey asked, "How many key performance indicators are monitored in your firm?" with response options ranging from "none" to "10 or more". The targets section asked about the design, integration, and feasibility of production targets. For example, the survey asked, "How easy or difficult is it in your firm for people to typically achieve their operational targets?" with answers ranging from "Possible to achieve without much effort" to "Only possible to achieve with extraordinary effort". Finally, the incentives section asked about bonus, promotion, and reassignment/dismissal practices. For example, the survey asked, "How were workers promoted in your firm?" with answers ranging from "no promotion" or "mainly on factors other than performance and ability, for example tenure or family connections" to "solely on performance and ability".

It is worth noting that, although the structure and nature of the questions in the two waves is different, it is possible to construct an indicator capturing – for each firm in two points in time – the adoption of performance-pay systems. Although we cannot properly identify the intensive margin (i.e., how important it is in the remuneration of the worker), we can nevertheless analyze the extensive one (i.e., its adoption or not) within firm in a panel setting.

3 Measure of managerial talent

Good manager characteristics reflect observable (and, ideally, observed) personal traits such as education, qualifications or previous experience. Moreover, they do comprise also unobserved ones, such as ability, charisma or leadership skills. Throughout the paper, we define talent as the individual portable and time-invariant contribution that a director brings to the TFP of the firm that she/he runs. Following this definition, the talent can be captured by the director's fixed effect in a regression that explores the determinants of firms' TFP. The idea that the talent of organization leaders can be measured by their fixed effects is present in the seminal work of Bertrand and Schoar (2003) on the role of top managers' identities on corporate policies of U.S. largest companies. More recently, it has been exploited in other settings by Graham et al. (2012), Lazear et al. (2015), Best et al. (2019) and Fenizia (2020).

3.1 Connected set of firms and directors' mobility

A two-way estimation of directors' and firms' fixed effects can be done, as shown by Abowd et al. (1999) (AKM, henceforth), insofar as there are directors who hold a seat in multiple firms, either in the same year or over time.¹⁴ The existence of such directors allows to separately estimate directors' and firms' fixed effects within each set of firms that are connected via directors' mobility and interlocking.

We observe around 470,000 directors (Table 1) of over 140,000 limited liability companies in our panel (Table 2). Nearly one fifth of the directors in our sample are observed in at least two different firms over the period considered in the analysis and we refer to such directors as "movers" in the remainder of the paper. The category of movers comprises individuals who are involved in the management of more than one firm either due to board interlocking or due to switching (i.e., the same manager moves from one firm to another over time). As shown in Figure 1, about 11% of the directors sit every year on boards of two different firms

¹⁴The AKM model was first used to separately estimate the effect of workers' and firms' time-invariant characteristics on individual wages. Card et al. (2013) provide a neat and detailed application of the AKM method to explain the drivers of the increasing wage inequality observed in West Germany.

(9% in two firms, nearly 3% in three or more firms). Moreover, every year about 6% of the directors exit from or enter into the board of a firm. In our sample period, about 15% of the directors move at least once (8% at least twice).¹⁵

In our analysis, we focus on the largest connected set of firms. This set includes 44% of firms with more than 20 employees and around 57% of their directors. Table 1 displays the observable characteristics of directors, separately for movers and non-movers, in the universe of the Italian limited liability companies (columns 1-2) and in the largest connected set of firms (columns 4-5). Columns 3 and 6 show whether differences between movers and non-movers within each of the two samples are statistically significant or not. Among the directors, nearly 80% are male and the corresponding figure is even larger among movers, indicating that men are more often involved in managing more than one firm. Furthermore, directors in the movers sample appear to be more often native Italians, born outside the province in which the firm is located and slightly older. The differences between mover and non-mover directors in the largest connected set are similar to those in the entire sample.

Table 2, columns 1 and 2 display the observable characteristics of firms in the largest connected set and other firms, respectively, while column 3 shows the test of statistical significance of the difference between the two samples. Firms in the largest connected set have 97 employees, on average, against the average size of 35 employees in other firms.¹⁶ Figure 2 illustrates the distribution of firms in the largest connected set by their size category. The left panel shows that the majority of firms in the largest connected set are small, representing 67% of the total (medium-sized and large firms, respectively, correspond to 27% and 6% of the total). Importantly, the right panel shows that the largest connected set is more representative of the universe for medium-sized and large firms, as it contains 63% and 86% of the total in these categories. Moreover, Table 2 reveals that firms in the largest connected set are shown that firms' networks, defined in terms of their directors' linkages, are seemingly more dense in Northern

¹⁵We provide additional information on the labor market of corporate directors in Appendix A.1.

¹⁶In the international comparison, the firms in our connected set are rather small: the average establishment size (arguably even smaller than firm size) in Bloom et al. (2019) is 180 employees.

and Central Italy. Finally, firms in the largest connected set are more often managed by a board of directors, on average composed of three components.

3.2 Estimation

The largest connected set consists of N firms and each firm *i* is observed over T_i years. We have therefore an unbalanced panel of $T = \sum_{i=1}^{i=N} T_i$ firm-year observations. In each year *t* a firm *i* is run by one or some among *J* directors, whose identities are known to us. This allows us to estimate the following high-dimensional two-way fixed effect model:

$$y = F\alpha + D\psi + X\beta + \varepsilon \tag{1}$$

y is a $T \times 1$ vector whose j-th element is the TFP of firm i in period t;¹⁷ F is a $T \times N$ matrix that collects firm dummies; D is a $T \times J$ matrix that collects directors dummies; X is a $T \times K$ matrix of year dummies; ε is the $T \times 1$ vector containing the error terms.

The OLS estimation of equation (1) provides a meaningful estimate of the coefficients ψ of interest as long as directors do not systematically sort into firms based on factors that are not observed by the econometricians and are thus included into the error term. As specification (1) features firm fixed effects, sorting based on companies time-invariant characteristics would not constitute a threat to identification. Following Card et al. (2013) we assume that the error term is composite and captures three forms of endogenous mobility: first, mobility patterns that depend on the idiosyncratic component of the firm-director match; second, mobility patterns based on the drift/trend component of firm TFP; third, mobility patterns that arise as a response to the transitory component of firm TFP. This amounts to assuming that directors do not sort into firms based on their comparative advantage. Moreover, directors should neither systematically leave or join firms whose productivity is declining or increasing over time, nor companies which experience a sharp change in their productivity. In sub-section 3.4 we discuss the validity of these assumptions in our setting.

¹⁷We use a measure of TFP that has been purged of sector-year and province-year fixed effects.

Figure 3 shows the distribution of directors' and firms' fixed effects, estimated based on specification (1). Both sets of fixed effect display a considerable dispersion, suggesting that there is substantial variation in directors' talent and firms' efficiency.¹⁸ After estimating individual fixed effects, we derive our measure of time-varying managerial talent available to the firm q as the average of its director fixed effects in any given year. We therefore compute:

$$q_{it} = \frac{\sum_{j \in J_{it}} \psi_j}{\sum_{j \in J_{it}} 1} \tag{2}$$

where J_{it} is the set of directors who run firm *i* in year *t*.

3.3 Managerial talent and firm productivity

In Table 3 we explore how much our measure of managerial talent available to the firm explains the variation in TFP. We begin by estimating a parsimonious model that only features sector-year and province-year fixed effects (column 1) and has a considerably low explanatory power. The model which adds firm fixed effects in column 2 has the adjusted R^2 equal to 0.54. In column 3 we further include the average observable board characteristics (i.e., gender, age, share of foreign-born and local-born director), but the explained variance of the model remains virtually unchanged. The inclusion of our measure of average managerial talent (q), in contrast, improves substantially the fit of the regression model (column 4): the adjusted R^2 goes up 0.64, corresponding to a 19% increase with respect to the previous specification. This finding suggests that variation in average board talent¹⁹ explains a significant

¹⁸Also the dependent variable, a firm TFP, is highly dispersed: in our data, the output of firms at the 90th percentile of TFP distribution is three times larger than that of firms at the 10th percentile (keeping constant the inputs).

¹⁹Table A.1 explores other measures of board talent, such as standard deviation of directors' fixed effects (column 2), talent of the most capable manager (column 3) and talent of executive directors (columns 4). We predict the probability of being an executive for individuals with an unknown executive-status based on observable characteristics of identified executive directors. We then use these probabilities to compute a weighted average of the directors' fixed effect to measure executives' talent. The results in Table A.1 indicate that a more homogeneous talent within the board positively affects the TFP; that the overall level of talent matters more than the talent of the single most capable manager; that the effect of executives on the firm TFP is comparable to the overall board talent effect (likely due to the fact that in small boards most directors have some executive powers).

portion of variation of the firm TFP, with an elasticity of around 0.75.

Table 4 explores heterogeneous effects of management talent. In line with the idea that more able managers are better at dealing with complex tasks, the effect of board talent increases both with firm size (column 1) and (non-linearly) with firm age (column 2). Column 3 shows that board talent has a stronger impact on firms in Northern Italy, while the effect decreases as we move to the Central and Southern parts of the country. Finally, columns 4 and 5, using a different and complementary perspective, show that managerial talent matters more in competitive environments. Indeed, the impact on the firm TFP is higher in urban and metropolitan areas than in rural ones – i.e., where agglomeration forces impose a tougher local competition – and in sectors more exposed to international trade. These findings mirror those obtained by Bloom and Van Reenen (2010), who show that product market competition is positively associated with the quality of management practices.

3.4 Validation and informative content of our measure of talent

Does our measure of talent meaningfully capture the portable and time-invariant ability of directors? To start with, we test the plausibility of the assumptions underlying the additive model of firms' efficiency and managerial ability postulated in equation (1). Based on a regression that controls for the interaction between firm fixed effects and board talent fixed effects, we document that endogenous sorting based on the idiosyncratic value of the match does not seem to be a relevant concern in our setting.²⁰ Furthermore, we examine the dynamics of TFP in the years preceding and following changes in the composition and the quality of the board and show that sorting based on the transitory component of firms' productivity does not seem to pose a serious threat in our setting. This exercise also reveals that the largest departure from the assumptions is related to sorting based on the trend component of productivity, i.e., more (less) talented directors appear to sort into firms whose performance is improving (deteriorating) over time. Although the data clearly exhibit a significant change in this trend upon the change in firm management quality, consistent with

 $^{^{20}}$ See the appendix A.2 for more details on the results of the validity checks proposed in Card et al. (2013).

the TFP-enhancing talent effect, we acknowledge that our measure of managerial talent may be somewhat overestimated in levels (this bias, however, is neutralized in the estimates of the interaction terms).

An alternative strategy to validate our measure of managerial talent is to examine its correlation with other observable measures of ability. To this end, we combine our data with the *Invind* survey information on firms' leaders and corporate performance. About 80% of the *Invind* sample firms are also present in the *Infocamere-Cerved* largest connected set. The combined *Infocamere-Cerved-Invind* sample includes around 6,200 firms. Compared to the overall sample, those firms are relatively larger (Table 2, column 4): the mean (median) size corresponds to 370 (90) workers. We use survey data to obtain *ex-ante* and *ex-post* indicators of ability, i.e., managers' educational attainment and forecasting precision of the firm's future performance.

The 2006 wave of the survey includes information on the level of education and on other anagraphic characteristics of the head of the companies. Exploiting these variables we identify the chief within the board of directors and examine whether his/her education is correlated with our measure of talent (Bloom et al., 2014). In particular, in Table 5 we regress managerial talent of the firm leader on his/her education attainment. The reference category comprises individuals with at most secondary school degree, while we estimate the coefficients of having a generic college degree, a college degree with some training in management and a college degree followed by a master or a PhD. The regression specification holds region, sector and firm size constant, and includes individual controls such as age or gender. The estimated coefficients indicate that while individuals with general college education are not characterized by higher talent, those with further training in management or post-graduate degrees are captured as high managerial talent leaders. This evidence is particularly compelling, given that the two exercises are based on completely different data and measurement approaches.

Being accurately informed about the company and the environment in which it operates is essential for making optimal corporate decisions, e.g., on the use of production factors and on the organization of the production process. A growing literature, in fact, has emphasized the importance of systematic errors in managerial forecasting in terms of investment decisions and firm performance (Malmendier and Tate, 2005; Ben-David et al., 2013). Using our survey data, we examine whether talented managers are associated with lower forecasting errors in the assessment of future firm performance. We define the forecasting error as the absolute difference between the predicted firms' revenues in the current year for the following year and the realized firm revenues in the following year (Ma et al., 2020). This variable is available in every survey wave, providing as with a firm-level panel for the period 2005-2018. We regress the absolute value of the prediction error on our measure of managerial talent, controlling for firm and year fixed effects. Table 6 shows that our measure of managerial talent is negatively associated with forecasting errors made. The effect is sizable: a one standard deviation change in managerial talent is associated to more than a one fifth standard deviation change in the forecasting error (column 1). The estimated impact is confirmed when we control for sectoral and local economic cycles (column 2) and for firm-specific demand shocks (column 3). Overall, this evidence strongly suggests that talented managers possess a more accurate set of information to assess the business environment in which the firm operates and to gear corporate decisions.

4 Complementarities

Modern production processes are complex and only rarely do they rely on using a single input. Typically, the final output is obtained from a combination of multiple factors; some of which are substitutes and can be used interchangeably, while others are complements. Following the definition in Brynjolfsson and Milgrom (2013), complementarities exist if the output produced by combining two or more economic factors in a production process exceeds what would have been otherwise generated through the use of the same factors in isolation. Generating such synergies allows to increase the "size of the pie" by merely combining different inputs rightly, without augmenting their quantity.

There is extensive evidence that differences in firm- and country-level productivity partly

reflect variation in management practices (Bloom and Van Reenen, 2010). However, little is known on whether good managerial practices matter *per se* or whether they are complementary to the talent of those who implement them (Syverson, 2011).²¹

To explore complementarities between managerial talent and managerial practices, we rely on the 2010 and 2019 waves of the *Invind* survey that include questions on the adoption of structured managerial practices within the firm. Table 7 shows the main descriptive statistics of these variables.²² We primarily rely on the more recent 2019 wave for two main reasons. First, the managerial score is built as in Bloom et al. (2019) and therefore our results square well within the existing literature. Second, we use a pre-determined measure of managerial talent in these regressions: we estimate director fixed effects using director-firm panel for the period 2005-2018 and relate it to the TFP and management score in 2019. Therefore, the forward-looking bias of our talent estimates is mitigated by the fact that only past firm performance is used to determine the ability of directors. We use the 2010 wave for robustness analysis and in the panel estimations on the use of performance-pay (as explained later).

We estimate the following cross-sectional regression specifications:

$$TFP_i = \alpha + \beta q_i + \gamma z_i + \theta q_i z_i + \rho x_i + \epsilon_i \tag{3}$$

where TFP_i is the firm's total factor productivity; q_i is the managerial talent; z_i is the managerial practices' score; x_i is a vector of firm-level controls (namely, the measure of firm quality, i.e. its fixed effect estimated in the largest connected set, sector, region and size fixed effects). The dependent variable and the key explanatory variables are all standardized, in order to guarantee an easier comparison of the relative size of the estimated effects and improve the readability of the results.

Table 8, column 1 presents the most parsimonious specification with the managerial score measure alone. The estimated coefficient confirms its positive effect on firm efficiency, in line

²¹See Bandiera et al. (2015) for an empirical analysis on the links between firm governance, incentives, and performance.

 $^{^{22}}$ See Figure A.7 in the Appendix for the questions asked in each wave.

with the existing literature. We augment the specification in column 2 with our measure of managerial talent; positive and significant coefficients on both variables show that, on average, each of them contributes positively to firms' efficiency. We include firm-level controls in column 3 and both variables continue to have a positive impact, although the coefficient of managerial score is no longer statistically significant and halves in magnitude. Finally, in column 4 we add the interaction between managerial score and talent: a significant and positive coefficient indicates the presence of synergies between the two factors: good practices matter more for firm productivity if they are adopted by talented managers.

We then examine in isolation each of the 8 items used to derive the managerial score to study for which features of structured organization the presence of talented manager is more relevant. In the interest of brevity, we plot the coefficients on the interaction term in Figure 4. Although all coefficients are positive, they are statistically significant in half of the cases. More precisely, the interaction term is significant for the number of monitored performance indicators, the frequency with which they are reviewed, the time frame of the operational targets and the criteria on which promotions are based.²³

One can imagine several reasons that may explain why some of the above-mentioned managerial tools appear to produce different results depending on how they are managed. For example, if more performance indicators are monitored, the manager may weigh those indicators to a different extent. Moreover, the manager may properly take into account internal and external factors that affect employees' performance and re-adjust the relative importance of each indicator. Finally, the manager, beyond correcting poor performance, needs to understand its causes: do they come from workers' ability or their effort? Incorrect

 $^{^{23}}$ We replicate the cross-sectional analysis using the 2010 wave as a robustness check. First, we condense the information contained in the three variables present in the 2010 wave that refer to firm-level organization and managerial practices into a composite indicator using the principal component analysis. Table A.2 shows that both the managerial score and managerial talent are significantly correlated with firm productivity and that there are complementarities between the two. Table A.3 replicates the analysis using the three single items separately. More specifically, we explore the role of team work (column 1), the adoption of performance-pay systems (column 2) and the involvement of lower hierarchical layers in the decision-making processes (column 3). The coefficients on the interaction terms show that the presence of complementarities is primarily due to synergies between managerial talent and performance-pay, while they seem to be absent for higher participation in decision-making and for the team-work.

diagnoses can lead to wrong organizational choices.

The design of targets is also fundamental, as they direct and guide workers' effort. Considering the dimensions included in the MOPS, there is a wide consensus that the targets should be challenging, yet attainable, and the time constraints should be fair and realistic. Indeed, if they are too easy, they will not boost performance, while if they are out of reach, the workers will be discouraged to even try to pursue them. Similarly, for what concerns the time frame, slack deadlines may slow performance, while too tight time schedules may lead to workers' burnout or to them giving up on the goal altogether. Nevertheless, managers may shape the effect of targets on firm performance in a number of ways. For example, goals should be designed to meet the wide objectives of the company and appear relevant within the job scope of the single worker. Moreover, targets should be aligned among the different departments within the company in order to ensure mutual support in implementing them and to prevent their conflict.

Finally, the adoption of performance-based remuneration schemes is widely viewed as a powerful tool to stimulate workers' effort. However, performance evaluation may be largely subjective. For certain tasks, there are quality dimensions that are both relevant and not trivial to measure; in turn, how a given performance is evaluated, may vary to a great extent from supervisor to supervisor. Moreover, more talented managers, when designing the incentive system, may be more capable in taking into account the heterogeneous effects of incentives across workers with different attitudes, willingness to take risks, relative self-assessment (Dohmen and Falk, 2011). Finally, when making promotion decisions, the managers might incorporate past performance measures to a different extent, as the ability to master the current job might not be necessarily correlated with the ability to do the subsequent one. Summing up, performance evaluation in promotion decisions is often based on subjective assessment by the managers who themselves may be less or more talented in recognizing relevant workers' ability and effort.

These considerations bring to the question on whether the extent of talent-practices complementarity is heterogeneous across firms. To shed light on this, in Table 9 we examine the role of managerial talent, managerial practices and their interaction in different subsamples of firms. First, we distinguish between firms operating in sectors exposed to international competition to a lesser or a greater extent. The underlying idea is that competitive environments likely place stronger pressure towards optimal corporate organization and management within firm. Since structured managerial practices may be relatively easy to imitate - in other words, the comparative firm advantage stemming from their use may decay quickly - one would expect that how the same practices are implemented should matter more when firms compete neck-to-neck. Second, we distinguish between firms in industries with high and low incidence of overtime work. This variable proxies for the relevance of workers' effort and we expect that complementarities between managerial talent and managerial practices are stronger in environments where effort is more important for the firms' output. While the coefficient on the interaction term is positive in all the specifications, it is statistically significant only in the subsample of firms more exposed to international competition and in those operating in effort-intensive industries.

The interpretation of our findings in causal terms requires that the key variables are not correlated with other relevant omitted variables. Although in the absence of a natural experiment we cannot rule out this possibility, we can nevertheless provide some robustness checks to indirectly support the causality hypothesis. There is a certain consensus in the literature that the relationship between managerial practices and firm performance may be interpreted in causal terms (Bloom et al., 2013; Giorcelli, 2019). We therefore mainly focus on our measure of managerial talent and present further evidence using the board talent measure estimated in the largest connected set that excludes firms belonging to the *Invind* sample ("leave-out" measure). This means that we build a measure for each director's talent based on their performance in firms outside the sample used to study complementarities. This attenuates the concern that directors' fixed effects may also capture time-varying characteristics of the firms or some idyosincratic features of the match between those firms and their directors. Despite this methodological advantage, the sample size diminishes by threefold and we are left with the "leave-out" board talent measure for only 350 firms. Table 10 replicates the regressions in Table 8 using this smaller set of firms. As above, managerial score and managerial talent are positively associated with firm productivity, although the impact of the former diminishes and becomes not significant when we include firm controls. Most important, the coefficient of the interaction term continues to be positive and statistically significant.

Last, but not the least, for the subset of questions that were asked both in 2010 and 2019, we build a two-period panel. Although this exercise is narrower in scope, it serves as an important robustness check allowing to argue that our results do not stem from the time-invariant firm-level confounders, jointly determining the presence of the complementarities and the firm performance. Indeed, the importance of "management panel data" has been greatly emphasized by Bloom and Van Reenen (2010), Syverson (2011)) and Bloom et al. (2019). For a sub-sample of 168 firms we jointly observe the TFP, our measure of board talent and the existence of performance-pay systems in both survey waves. For this sample we run the following panel regressions which exploit the within-firm variation in our variables of interest:

$$TFP_{it} = \alpha + \beta q_{it} + \gamma z_{it} + \theta q_{it} z_{it} + \rho_i + \tau_t + \epsilon_{it}$$

$$\tag{4}$$

where TFP_{it} is the firm's total factor productivity in 2010 and 2019; q_{it} is the time-varying measure of managerial talent; z_{it} is an indicator equal to 1 if the firm has a performance-pay system and is, again, time-varying; ρ_i and τ_i are firm- and year-fixed effects, respectively.

Table 11 shows that firms that adopt incentives related to performance between 2010 and 2019 increase their TFP (column 1). The positive impact of the adoption of these schemes is confirmed when we control for managerial talent (column 2), although the point estimate is slightly lower. Moreover, an increase in managerial talent is also associated with a significant improvement in firm productivity. Finally, the coefficient on the interaction term is positive and statistically significant (column 3), revealing the presence of complementarities between *acquisition* of structured management practices and *hiring* of talented managers. In quantitative terms, the coefficients estimated in the last column show that the adoption

of a performance-pay system is associated to a nearly 9% increase in the TFP, while moving from a firm at the 25th to one at the 75th percentile of board talent is associated with a 21% increase of the TFP. Interestingly, such increase in managerial talent leads to a further 8% increase in the TFP if associated with the adoption of performance-related workforce remuneration.²⁴ Not surprisingly, the presence of good managers enhances the efficiency of those practices which require the direct involvement of top management of the company: performance-pay is more effective in firms where top managers who set the employee objectives and assess their performance are talented.

5 Concluding remarks

This paper explores the presence of complementarities between talended management and structured managerial practices in determining firm-level productivity. The analysis exploits a novel and rich dataset on Italian limited liability companies to derive a measure of timeinvariant and portable component of talent for individuals responsible for the key corporate decisions. We present evidence that managerial talent matters on its own, contributing significantly to the firm's TFP.

We also show that individuals whom we identify as talented managers more often possess adequate education or training, and are better in precisely forecasting the firm's performance. This suggests that directors' fixed effects reflect (albeit noisily) true managerial acumen and talent.

Most importantly, we relate our measure of managerial talent to the detailed survey data on management practices to better understand their role separately and jointly in determining firm's performance. The data reveals synergies between managerial ability and managerial practices. While managerial talent and the use of good managerial practices do boost firm productivity on their own, there is evidence of complementarities between them.

 $^{^{24}}$ See Lazear (2000) for a seminal paper on the relationship between incentive-based pay and productivity. Using a case study, he finds that the adoption of incentive based pay was associated to a 44% increase in the output per worker.

While we admit our inability to make strong causal claims, our findings are new and intuitive. Moreover, they are confirmed when we account for potential time-varying characteristics of the firms affecting our measures of managerial talent and we control for firm fixed-effects (capturing all time-invariant heterogeneity at the firm level).

The current literature still faces the chicken-egg question regarding manager selection and managerial practices. On the one hand, structured managerial practices may facilitate the match between firms and talented managers. On the other hand, the causality may run in the other direction, if talented managers happen to implement better organizational and managerial practices. Even though both explanations remain plausible, our novel evidence is interesting because it documents the positive interaction between the two factors in affecting firm performance: more talented managers are better able to make use of structured managerial practices.

Drawing policy implications in the field of corporate governance is difficult, because the manager selection and the adoption of certain organizational practices are decisions ultimately taken within the firm. The public interventions may, nevertheless, affect the general environment in which firms operate. We find that higher education achievements are associated with higher managerial talent. The policy maker may, therefore, invest in human capital, thus improving – quantitatively and qualitatively – the talent pool from which managers are selected. We also find that managerial talent is more important in more competitive markets. Favoring market competition, therefore, may seem a powerful tool to increase management efficiency within the firm and across firms, e.g., by removing frictions in talent allocation within the economy. More generally, promotion of organizational and managerial culture may help spreading productivity-boosting practices in the private sector.

References

- Abowd, J. M., F. Kramarz, and D. N. Margolis (1999). High wage workers and high wage firms. *Econometrica* 67(2), 251–333.
- Ackerberg, D. A., K. Caves, and G. Frazer (2015). Identification properties of recent production function estimators. *Econometrica* 83(6), 2411–2451.
- Adams, R. B., A. C. Akyol, and P. Verwijmeren (2018). Director skill sets. Journal of Financial Economics 130(3), 641–662.
- Andrews, M. J., L. Gill, T. Schank, and R. Upward (2008). High wage workers and low wage firms: negative assortative matching or limited mobility bias? *Journal of the Royal Statistical Society* 171(3), 673–697.
- Baltrunaite, A., E. Brodi, and S. Mocetti (2019). Assetti proprietari e di governance delle imprese italiane: nuove evidenze e effetti sulla performance delle imprese. Bank of Italy Occasional Paper (514).
- Bandiera, O., L. Guiso, A. Prat, and R. Sadun (2015). Matching firms, managers, and incentives. *Journal of Labor Economy* 33(3), 623–681.
- Bandiera, O., S. Hansen, A. Prat, and R. Sadun (2020). Ceo behavior and firm performance. Journal of Political Economy 128(4), 1325–1369.
- Ben-David, I., J. R. Graham, and C. R. Harvey (2013). Managerial miscalibration. Quarterly Journal of Economics 128(4), 1547–1584.
- Bender, S., N. Bloom, D. Card, J. Van Reenen, and S. Wolter (2018). Management practices, workforce selection, and productivity. *Journal of Labor Economics* 36(S1), S371–S409.
- Bennedsen, M., K. M. Nielsen, F. Pérez-González, and D. Wolfenzon (2007). Inside the family firm: The role of families in succession decisions and performance. *Quarterly Journal of Economics* 122(2), 647–691.

- Bernile, G., V. Bhagwat, and S. Yonker (2018). Board diversity, firm risk, and corporate policies. *Journal of Financial Economics* 127(3), 588–612.
- Bertrand, M. and A. Schoar (2003). Managing with style: The effect of managers on firm policies. *Quarterly Journal of Economics* 118(4), 1169–1208.
- Best, M., J. Hjort, and D. Szakonyi (2019). Individuals and organizations as sources of state effectiveness. *working paper*.
- Bloom, N., E. Brynjolfsson, L. Foster, R. Jarmin, M. Patnaik, I. Saporta-Eksten, and J. Van Reenen (2019). What drives differences in management practices? *American Economic Review* 109(5), 1648–1683.
- Bloom, N., B. Eifert, A. Mahajan, D. McKenzie, and J. Roberts (2013). Does management matter? evidence from india. *Quarterly Journal of Economics* 128(1), 1–51.
- Bloom, N., R. Lemos, R. Sadun, D. Scur, and J. Van Reenen (2014). The new empirical economics of management. *Journal of the European Economic Association* 12(4), 835–876.
- Bloom, N. and J. Van Reenen (2007). Measuring and explaining management practices across firms and countries. *Quarterly Journal of Economics* 122(4), 1351–1408.
- Bloom, N. and J. Van Reenen (2010). Why do management practices differ across firms and countries? *Journal of Economic Perspectives* 24(1), 203–224.
- Brynjolfsson, E. and P. Milgrom (2013). Complementarity in organizations. In R. Gibbons and J. Roberts (Eds.), *Handbook of Organizational Economics*. Princeton University Press.
- Card, D., J. Heining, and P. Kline (2013). Workplace heterogeneity and the rise of west german wage inequality. *Quarterly Journal of Economics* 128(3), 967–1015.
- Cornwell, C., I. M. Schmutte, and D. Scur (2021). Building a productive workforce: the role of structured management practices. *Management Science*, forthcoming.

- Dohmen, T. and A. Falk (2011, April). Performance pay and multidimensional sorting: productivity, preferences, and gender. *American Economic Review* 101(2), 556–90.
- Fenizia, A. (2020). Managers and productivity in the public sector. *working paper*.
- Gibbons, R. and R. Henderson (2013). What do managers do? exploring persistent performance differences among seemingly similar enterprises. In R. Gibbons and J. Roberts (Eds.), *Handbook of Organizational Economics*. Princeton University Press.
- Giorcelli, M. (2019). The long-term effects of management and technology transfers. American Economic Review 109(1), 121–152.
- Graham, J. R., S. Li, and J. Qiu (2012). Managerial attributes and executive compensation. *Review of Financial Studies* 25(1), 144–186.
- Ichniowski, C., K. Shaw, and G. Prennushi (1997). The effects of human resource practices on manufacturing performance: A study of steel finishing lines. *American Economic Review* 87(3), 291–313.
- Kaplan, S. N., M. M. Klebanov, and M. Sorensen (2012). Which ceo characteristics and abilities matter? *Journal of Finance* 67(3), 973–1007.
- Lazear, E. P. (2000). Performance pay and productivity. American Economic Review 90(5), 1346–1361.
- Lazear, E. P. and K. L. Shaw (2007). Personnel economics: The economist's view of human resources. Journal of Economic Perspectives 21(4), 91–114.
- Lazear, E. P., K. L. Shaw, and C. T. Stanton (2015). The value of bosses. Journal of Labor Economics 33(4), 823–861.
- Levinsohn, J. and A. Petrin (2003). Estimating production functions using inputs to control for unobservables. *Review of Economic Studies* 70(2), 317-341.

- Lieberman, M. B., L. J. Lau, and M. D. Williams (1990). Firm-level productivity and management influence: A comparison of us and japanese automobile producers. *Management Science* 36(10), 1193–1215.
- Ma, Y., T. Ropele, D. Sraer, and D. Thesmar (2020). A quantitative analysis of distortions in managerial forecasts. *NBER Working Papers* (26830).
- Malmendier, U. and G. Tate (2005). Ceo overconfidence and corporate investment. *Journal* of Finance 60(6), 2661–2700.
- Melitz, M. J. (2003). The impact of trade on intra-industry reallocations and aggregate industry productivity. *Econometrica* 71(6), 1695–1725.
- Syverson, C. (2004). Product substitutability and productivity dispersion. *Review of Economics and Statistics* 86(2), 534–550.
- Syverson, C. (2011). What determines productivity? Journal of Economic Literature 49(2), 326-365.

Tables and figures

	(1)	(2)	(3)	(4)	(5)	(6)	
	unive	universe of firms			connected set		
	Non-movers	Movers	Δ	Non-movers	Movers	Δ	
Female	0.249	0.143	0.107***	0.231	0.131	0.100***	
Foreign-born	0.105	0.084	0.021^{***}	0.109	0.082	0.027^{***}	
Age	49.623	51.277	-1.655^{***}	50.187	51.733	-1.546^{***}	
Local	0.693	0.619	0.074^{***}	0.666	0.603	0.063^{***}	
Shareholder	0.347	0.248	0.099^{***}	0.220	0.198	0.023^{***}	
Talent				-0.005	-0.003	-0.002	
Share	0.811	0.189		0.734	0.266		
N		472,067			269,927		

Table 1: Descriptive statistics on directors

Notes: Data are drawn from the combined *Infocamere-Cerved* sample, including the universe of limited companies with more than 20 employees in the private non-financial sector in the years 2005-2018. Columns (1) and (2) report mean values for movers and non-movers for the full sample, while columns (4) and (5) report the same figures for the subsample of firms included in the largest connected set; Δ indicates the corresponding difference in means; N represents the total number of directors in the period considered.

	(1)	(2)	(3)	(4)
	other firms	connected set	Δ	Invind sample
# employees	34.660	97.057	-62.397***	370.000
Firm age	14.231	18.107	-3.876***	30.800
# directors	1.874	3.229	-1.356***	4.430
% manufacturing	0.355	0.321	0.034^{***}	0.603
% South	0.288	0.146	0.142^{***}	0.252
TFP	-0.060	-0.040	-0.020***	0256
Share	0.559	0.441		
N	14	4,632		6,224

Table 2: Descriptive statistics on firms

Notes: Data are drawn from the combined *Infocamere-Cerved* sample, including the universe of limited companies with more than 20 employees in the private non-financial sector in the years 2005-2018, in the first two columns and from the combined *Infocamere-Cerved-Invind* sample in the last column. Columns (1) and (2) report mean values for firms outside the largest connected set and within it, respectively; Δ indicates the corresponding difference in means; the last column reports mean values for the subsample of the connected firms included in the *Invind* survey. N represents the total number of firms in the period considered.

Dependent variable:			T	FP		
	(1)	(2)	(3)	(4)	(5)	(6)
Industry \times Year FE	Yes	Yes	Yes	Yes	Yes	Yes
Province \times Year FE	Yes	Yes	Yes	Yes	Yes	Yes
Firm FE		Yes	Yes	Yes	Yes	Yes
Board characteristics			Yes	Yes	Yes	Yes
Board talent				0.761^{***}		
				(0.005)		
Board FE					Yes	Yes
Board FE \times Firm FE						Yes
4.14.752	0.004					0.004
Adj-R ²	0.024	0.536	0.536	0.637	0.658	0.634
Ν	479,038	479,038	479,038	479,038	479,038	479,038

Table 3: Board talent and firm productivity: analysis of the variance

Notes: Data are drawn from the combined *Infocamere-Cerved* sample in the years 2005-2018. Panel with fixed effects. The table shows how much of the variation of firm TFP is explained by: industry- and province-year FEs (column 1); firm FEs (column 2); the observable characteristics of the board of directors (column 3); the talent of the board (measured as the average of director fixed effects at the firm-year level), both as a continuous variable (column 4) and as a set of fixed effects corresponding to its centiles (column 5); the interaction between firm fixed effects and board talent fixed effects (column 6). Standard errors clustered at the firm level in parentheses; *** p<0.01, ** p<0.05, * p<0.1.

Dependent variable:			TFP		
	(1)	(2)	(3)	(4)	(5)
Board talent (BT)	0.745^{***} (0.005)	0.739^{***} (0.007)	$\begin{array}{c} 0.823^{***} \\ (0.007) \end{array}$	0.731^{***} (0.009)	$\begin{array}{c} 0.685^{***} \\ (0.010) \end{array}$
BT \times medium	0.054^{***} (0.008)				
$BT \times large$	0.109 ^{***} (0.022)				
BT \times age 10-30		0.005 (0.007)			
BT \times age 30+		0.044^{***} (0.009)			
$BT \times Centre$			-0.104^{***} (0.013)		
$BT \times South$			-0.193^{***} (0.012)		
BT \times urban area			``	0.037^{***} (0.013)	
BT \times metropolitan area				0.048^{***} (0.013)	
BT \times mid export					0.073^{***} (0.013)
BT \times high export					0.133^{***} (0.014)
Firm FE	Yes	Yes	Yes	Yes	Yes
Industry \times Year FE	Yes	Yes	Yes	Yes	Yes
Province \times Year FE	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.677	0.677	0.678	0.677	0.677
N	479,038	479,038	479,038	756,283	479,038

 Table 4: Heterogeneous effects

Notes: Data are drawn from the combined *Infocamere-Cerved* sample in the years 2005-2018. Panel with fixed effects. The dependent variable is firm TFP. Board talent is interacted with various firm characteristics. The residual categories are small firm (column 1), aged less than 10 (column 2), North (column 3), rural area (column 4) and low export sectors (column 5). Standard errors clustered at the firm level in parentheses; *** p < 0.01, ** p < 0.05, * p < 0.1.

Dependent variable:	Managerial talent			
	(1)	(2)	(3)	
College	0.039	0.049	0.070	
	(0.052)	(0.053)	(0.054)	
+ training in management	0.287^{*}	0.303^{**}	0.283^{*}	
	(0.148)	(0.145)	(0.145)	
+ master/PhD	0.390^{**}	0.407^{***}	0.459^{***}	
	(0.154)	(0.156)	(0.163)	
Individual controls		Yes	Yes	
Firm controls			Yes	
\mathbb{R}^2	0.006	0.008	0.041	
Ν	1,585	1,585	1,585	

Table 5: Managerial talent and education achievements

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the the 2006 wave. OLS cross-section regression. The dependent variable is managerial talent of the firm leader while the explanatory variables are her/his education achievements: the reference category is "no college", while the estimated dummies refer to "college degree", "college degree plus a training in management of at least three months" and "college degree plus a master or a PhD". Individual controls include age and gender; firm controls include region, sector and size fixed effects. Standard errors clustered at the individual level in parentheses; *** p < 0.01, ** p < 0.05, * p < 0.1.

Dependent variable:	Forecasting error				
	(1)	(2)	(3)		
Board talent	-0.235^{***} (0.026)	-0.231^{***} (0.025)	-0.166^{***} (0.023)		
Log of revenues			-0.520^{***} (0.048)		
Firm FE	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes		
Industry \times Year FE		Yes	Yes		
Region \times Year FE		Yes	Yes		
\mathbb{R}^2	0.376	0.402	0.415		
Ν	26,745	26,745	26,745		

Table 6: Managerial talent and forecasting error

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample in the years 2005-2018. Panel with fixed effects. The dependent variable is the managerial forecast error defined as the absolute value of the percentage difference between the predicted revenues in the current year for the subsequent year and realized revenues in the subsequent year. Standard errors clustered at the firm level in parentheses; *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)
	Mean	Standard deviation	Min	Max
		2010 wave		
Managerial score	0.010	1.414	-1.663	3.264
Team Work	1.996	1.109	1.000	4.000
Performance pay	1.992	1.124	1.000	4.000
Decision-making	2.060	0.955	1.000	4.000
		2019 wave		
Managerial score	0.510	0.187	0.000	0.979
Monitoring score	0.586	0.237	0.000	1.000
Targeting score	0.613	0.275	0.000	1.000
Incentives score	0.365	0.262	0.000	1.000

Table 7: Descriptive statistics on organization and managerial practices

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample. In the 2010 wave the managerial score is the first principal component of the following managerial practices: presence of team work, adoption of performance-pay and involmment of lower hierarchical levels in the decision-making process. In the 2019 wave the managerial score is computed following Bloom et al. (2019) on the basis of 8 scores that can be aggregated in three groups: monitoring activity, targets and incentives. See the Appendix for the questions included in the survey and the associated scores.

Dependent variable:				
	(1)	(2)	(3)	(4)
Managerial score (MS)	0.051^{**} (0.024)	0.045^{*} (0.026)	0.020 (0.029)	0.020 (0.028)
Board talent (BT)		0.746^{***} (0.055)	0.756^{***} (0.056)	0.765^{***} (0.060)
$MS \times BT$		× /		0.081^{**} (0.039)
Firm controls			Yes	Yes
\mathbb{R}^2	0.002	0.326	0.330	0.336
Ν	$1,\!683$	1,078	1,078	1,078

Table 8: Complementarities: management talent and managerial score

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the 2019 wave. OLS cross-section regression. The dependent variable is firm TFP while the explanatory variable is the managerial score computed following Bloom et al. (2019). Firm controls include firm fixed effects - those estimated using the two-way fixed effect model described in specification (1) - and sector, region and size fixed effects. Standard errors clustered at the firm level in parentheses; *** p<0.01, ** p<0.05, * p<0.1.

Dependent variable:	TFP				
	export orie	entation	effort inte	ensity	
	Low	High	Low	High	
	(1)	(2)	(3)	(4)	
Managerial score (MS)	0.001	0.063	0.033	0.013	
	(0.037)	(0.052)	(0.044)	(0.037)	
Board talent (BT)	0.841^{***}	0.666^{***}	0.791^{***}	0.729^{***}	
	(0.087)	(0.072)	(0.076)	(0.081)	
$MS \times BT$	0.070	0.087^{*}	0.033	0.158***	
	(0.064)	(0.047)	(0.048)	(0.060)	
Firm controls	Yes	Yes		Yes	
\mathbb{R}^2	0.412	0.312	0.368	0.374	
Ν	567	507	548	527	

Table 9: Complementarities: heterogeneous effects

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the 2019 wave. OLS cross-section regression. The dependent variable is firm TFP while the explanatory variable is the managerial score computed following Bloom et al. (2019). Firm controls include firm fixed effects - those estimated using the two-way fixed effect model described in specification (1) - and sector, region and size fixed effects. Standard errors clustered at the firm level in parentheses; *** p<0.01, ** p<0.05, * p<0.1.

Dependent variable:	TFP				
	(1)	(2)	(3)	(4)	
Managerial score (MS)	0.094^{*} (0.056)	0.102^{*} (0.056)	0.057 (0.064)	0.060 (0.064)	
Board talent (BT)	× ,	0.165^{***} (0.059)	0.160^{**} (0.063)	0.173^{***} (0.063)	
$MS \times BT$, , , , , , , , , , , , , , , , , , ,		0.071^{*} (0.041)	
Firm controls				Yes	
\mathbb{R}^2	0.065	0.085	0.157	0.162	
IN	390	300	390	300	

Table 10: Complementarities: leave-out robustness check

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the 2019 wave. OLS cross-section regression. The dependent variable is firm TFP while the explanatory variable is the managerial score computed following Bloom et al. (2019). Firm controls include firm fixed effects - those estimated using the two-way fixed effect model described in specification (1) - and sector, region and size fixed effects. Standard errors clustered at the firm level in parentheses; *** p<0.01, ** p<0.05, * p<0.1.

	P		
Dependent variable:		TFP	
	(1)	(2)	(3)
Incentives related to performance (I)	0.103^{**} (0.041)	0.084^{**} (0.041)	0.086^{**} (0.040)
Board talent (BT)		2.206^{***} (0.374)	1.900^{***} (0.308)
$I \times BT$			0.731^{***} (0.224)
Firm FEs	Yes	Yes	Yes
Year FEs	Yes	Yes	Yes
\mathbb{R}^2	0.785	0.861	0.870
Ν	336	336	336

Table 11: Complementarities: panel analysis

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the 2010 and 2019 waves. Panel with fixed effects. The dependent variable is firm TFP while the explanatory variable is an indicator for the presence of performance-related pay systems. The model includes firm and year fixed effects, thus exploiting within firm variation in TFP, board talent and incentives. Standard errors clustered at the firm level in parentheses; *** p < 0.01, ** p < 0.05, * p < 0.1.



Figure 1: Extent of interlocking and switching among directors

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, including the universe of limited companies with more than 20 employees in the private non-financial sector in the years 2005-2018. The left panel shows the extent of interlocking, i.e. the distribution of directors by the number of boards (of different firms) on which they seat in the same year; the right panel shows the extent of switching, i.e. the distribution of directors time) over the period 2005-2018.



Figure 2: Distribution of firms in the connected set by size

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, including the universe of limited companies with more than 20 employees in the private non-financial sector in the years 2005-2018. The left panel shows the distribution of firms in the connected set by size; the right panel shows the share of firms in the connected set with respect to the universe by size. Small firms have from 20 to 50 employees, medium-sized firms have up to 250 employees while large firms have more than 250 employees.



Figure 3: Distribution of firms' and directors fixed-effects

Notes: Data are drawn from the combined *Infocamere-Cerved* sample. Distribution of firms' and directors' fixed effects estimated through the two-way fixed effect model. Both variables are standardized.



Figure 4: Complementarities: management talent and managerial practices (2019)

Notes: Each point represents the estimated coefficient of the interaction term between managerial talent and managerial practices, for different managerial items. Vertical bandsrepresent + / - 1.66 time the standard error for each point estimate. Data are drawn from the combined *Infocamere-Cerved-Invind* sample. The score for each items capture, respectively, "what happens when a problem in the production process arises", "how many key performance indicators are monitored", "how frequently are key performance indicators typically reviewed", "what is the time frame of operational targets", "how easy or difficult is it for workers to typically achieve their operational targets", "what are performance bonuses usually based on", "what is the primary way workers are promoted", "when is an under-performing worker usually reassigned or dismissed". See Bloom et al. (2019) for more details on the questions and the relative scores.

Appendix

A.1 Stylized facts on the labour market of directors

As shown in Section 3, many directors sit on the board of multiple firms over the period 2005-2018. Directors tend to move across firms that are "close" from both a geographical and sectoral point of view (Figure A.1).²⁵ We find that 45% of the moves (interlocking and/or switching) occur between firms that are located within the same municipality, while around 22% of the moves occur between firms more than 100 kilometers away. Similarly, directors tend to sit on the boards of firms belonging to the same sector of activity: the likelihood that the two firms belong to the same section (alphabetical classification), division (2 digit numerical classification) or group (3 digit numerical classification) of the NACE classification of economic activities are, respectively, around 56%, 39% and 32%. These probabilities are significantly larger than those that would be recorded by observing a random shift from one sector to another.²⁶

A.2 Validity checks on the AKM model

As explained in Section 3, specification (1) relies on the assumption that directors do not systematically sort into firms based on factors that enter in the error term. In this section, we test three patterns of endogenous mobility that would violate this assumption. To this end, we perform the validity checks proposed in Card et al. (2013).

²⁵To construct measures of geographical or sectoral distance we need the manager to be present in at least two firms, one of origin and one of destination, simultaneously (as in the case of interlocking) or sequentially (as in the case of switching). To simplify the analysis we have considered all the cases with interlocking equal to two (for the cross-sectional component) and all the cases in which the administrator leaves a company and, in the following year, enters another (for the longitudinal component).

 $^{^{26}}$ An alternative way to capture sectoral proximity is to examine if the move of the director takes place between firms belonging to the same production chain. Using the input-output matrices we consider, for each combination of branches of economic activity, the average between the fraction of output of the branch of origin used as input in the branch of destination and the fraction of output of the destination branch used as input in the branch of origin. This figure, that captures how much two branches are integrated in the same production chain, is equal to 11% for the moves that we observe, 6 times larger than the simple average obtained from a random move.

First, we consider sorting based on the idiosyncratic component of the match. If this form of endogenous mobility is not relevant, a fully saturated model that features the interaction between the fixed effects of directors and firms should not have a significantly larger explanatory power than our baseline model. To test this, in column 6 of Table 3 we regress a firm TFP on the interaction between fixed effects for the average managerial talent q (obtained after discretizing the continuous measure into centiles) and firm fixed effects.²⁷ The adjusted R^2 of this model is not larger than that of the additive model in column 5, suggesting that match-specific effects should not have a first-order relevance in determining the sorting of directors into companies.

Moreover, if match-specific effects are not relevant the additive model should not deliver abnormally large residuals. Figure A.2 plots mean residuals in each of the 100 cells defined by the interaction of deciles of firms' and directors' fixed effects estimated in specification (1): the mean residuals in each cell are small and never exceeding the rule-of-thumb value of 0.02.

Furthermore, if this form of endogenous mobility is not important, we should observe that productivity gains experienced by companies that improve their managerial talent are roughly symmetric to productivity losses undergone as a result of a decline of similar extent in managerial talent.²⁸ To check for this, we focus on a balanced panel of firms that (i) change at least one director in year-to-event 0 and (ii) do not experience any other significant change - in terms of management quality - in the 3 years before the event and in the 3 years following it. We classify these companies into 9 groups, based on the terciles of managerial talent of the old board and the new board. Figure A.3 plots the evolution of TFP from year-to-event -3 to year-to-event 2 for firms whose old director/board belongs to the bottom or top tercile of managerial talent. The figure shows no change in TFP if changes in the composition of

²⁷In column 5 we estimate the same model as in column 4, substituting the continuous measure of board talent with indicators for each centile of the corresponding discretized measure. Although the discrete variable has less informative value than the continuous variable, the use of fixed effects allows capturing potential non-linearities in the relationship between board talent and firm TFP. The fit of the model is however only marginally affected by this change.

²⁸On the other hand, if match effects are relevant, gains would be larger than losses, as directors would systematically sort into companies where they have a better match.

the board do not result in a change in managerial talent (i.e., for transitions of the type 1 to 1 or 3 to 3). Focusing on changes of intermediate intensity (i.e., for transitions of the type 1 to 2 and 3 to 2), TFP is fairly flat before year-to-event 0, while it starts to increase (decrease) when a higher (lower) talented board takes over. Extreme changes in the managerial talent available to the firm (i.e., for transition of the type 1 to 3 and 3 to 1) are associated with larger changes in TFP from year 0, although these positive (negative) extreme changes are also preceded by increasing (declining) trends in TFP. However, these (mild) pre-trend do not seem to explain entirely the jump observed when extreme changes in board talent occurs, as shown in Figure A.4 that plots the observed pattern of the TFP and that predicted extrapolating from the trend observed before the change in the board quality. Finally, Figure A.5 plots the overall change in TFP (between year-to-event -3 and year-to-event 2) for downward movers against that of upward movers making the opposite change in managerial talents.²⁹ Dots are close to the the -45 degree line, indicating that TFP gains and losses for companies that experience opposite changes in board talent are roughly symmetric.

Second, we turn our attention to endogenous mobility based on the trend component of TFP. If sorting based on trends was not important, TFP should display a flat dynamic before a director leaves or joins the firm. As commented in the above paragraph, this appears to be the case when considering board changes that involve little or medium changes in the level of managerial talent available to the firm. On the other hand, this assumption seems less plausible when larger changes in managerial talent occur. A consequence of this is that we could be overstating the impact of managerial quality on TFP. However, Figure A.3 also shows that, even in these types of transitions, the changes in TFP before a new director of different talent joins/leaves the firm are lower than those observed after: the evident change in the slope suggests therefore that managerial talent still has an effect. Stated differently, the kink in the TFP can be attributed to the variation in board talent.

²⁹This figure also includes transitions from the middle to either the bottom or the top tercile of managerial talent.

Third, we examine endogenous mobility related to the transitory component of TFP. If this type of sorting was relevant, we should observe dips or spikes in TFP just before the change in the composition of the board. Such patterns do not emerge from Figure A.3, suggesting that this type of endogenous mobility is likely not of first-order relevance.

Last, more talented directors tend to be in firms with higher total factor productivity, as shown in Figure A.6. The joint distribution of firms and directors fixed effects, in contrast, highlights the presence of the negative assortative matching, in line with what has been found in other studies examining workers-firms matching processes and possibly due to standard estimation error (Andrews et al., 2008). Yet, it is worth noting that this is not problematic for the estimation of the managers fixed effects, as the model already absorbs all time-invariant firm characteristics.

A.2.1 Supplementary figures and tables



Figure A.1: Geographical and sectorial distance of "moves" between firms

Notes: Data are drawn from the combined *Infocamere-Cerved* sample. We consider as moves both the presence in the board of two different firms in the same year and the switch from one firm to another across time.

Dependent variable:		TF	Р	
	(1)	(2)	(3)	(4)
Board talent (mean)	0.761^{***} (0.005)	0.761^{***} (0.005)		
Board talent (sd)		-0.009^{**} (0.004)		
Board talent (max)			0.424^{***} (0.006)	
Executives talent (mean)				0.741^{***} (0.005)
Firm FE	Yes	Yes	Yes	Yes
Industry \times Year FE	Yes	Yes	Yes	Yes
Province \times Year FE	Yes	Yes	Yes	Yes
Board characteristics	Yes	Yes	Yes	Yes
$\mathrm{Adj}\text{-}\mathrm{R}^2$	0.637	0.637	0.569	0.630
Ν	479,038	479,038	479,038	479,038

Table A.1: Board talent and firm productivity: various measures of board talent

Notes: Data are drawn from the combined *Infocamere-Cerved* sample in the years 2005-2018. Panel with fixed effects. The table shows how much of the variation of firm TFP is explained by industry- and province-year FEs, firm FEs, the observable characteristics of the board of directors and the talent of the board. In column 1 board talent is the average of directors fixed effect at the firm-year level; in column 2 it is captured by both the average and the standard deviation of directors fixed effects; in column 3 board talent is the highest among directors' fixed effects; in column 4 it is the average of executive fixed effects. Standard errors clustered at the firm level in parentheses; *** p < 0.01, ** p < 0.05, * p < 0.1.

Dependent variable:		TFP		
	(1)	(2)	(3)	(4)
Managerial score (MS)	0.110^{***} (0.036)	0.059^{**} (0.029)	0.056^{*} (0.029)	0.031 (0.031)
Board talent (BT)		0.768^{***} (0.063)	0.766^{***} (0.064)	0.738^{***} (0.066)
$MS \times BT$		()	0.074^{*}	0.072^{*}
Firm controls			(0.012)	Yes
\mathbb{R}^2	0.012	0.409	0.415	0.459
1 N	000	010	010	010

Table A.2: Complementarities: management talent and managerial score (2010)

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the 2010 wave. OLS cross-section regression. The dependent variable is firm TFP while the explanatory variable is the managerial score (the first principal component of the following managerial practices: presence of team work, adoption of performance-pay and involmment of lower hierarchical levels in the decision-making process). Firm controls include firm fixed effects - those estimated using the two-way fixed effect model described in specification (1) - and sector, region and size fixed effects. Standard errors clustered at the firm level in parentheses; *** p<0.01, ** p<0.05, * p<0.1.

Dependent variable:		TFP	
	(1)	(2)	(3)
Board talent (BT)	0.740***	0.738***	0.741^{***}
	(0.065)	(0.062)	(0.065)
Team work (TW)	0.023		
	(0.032)		
$TW \times BT$	-0.023		
	(0.049)		
Performance-pay (PP)		0.058^{*}	
		(0.031)	
$PP \times BT$		0.183^{***}	
		(0.046)	
Decision-making (DM)			-0.007
			(0.031)
$DM \times BT$			0.028
			(0.043)
Firm controls	Yes	Yes	Yes
\mathbb{R}^2	0.454	0.487	0.454
Ν	678	678	678

Table A.3: Complementarities: management talent and managerial practices (2010)

Notes: Data are drawn from the combined *Infocamere-Cerved-Invind* sample, using the 2010 wave. OLS cross-section regression. The dependent variable is firm TFP while the explanatory variables are the following managerial practices: presence of team work, adoption of performance-pay and involmment of lower hierarchical levels in the decision-making process. Firm controls include firm fixed effects - those estimated using the two-way fixed effect model described in specification (1) - and sector, region and size fixed effects. Standard errors clustered at the firm level in parentheses; *** p < 0.01, ** p < 0.05, * p < 0.1.



Notes: Data are drawn from the combined Infocamere-Cerved sample. Figure shows mean residuals from model (1) on the largest connected set with cells defined by deciles of board talent, interacted with deciles of estimated firm fixed effects.



Notes: The figure plots the evolution of TFP from year-to-event -3 to year-to-event 2 on a balanced subset of firms that (i) change at least one director in year-to-event 0 and (ii) remain in the same tercile of board talent both in the 3 years before the event and in the following 3 years.



Figure A.4: Predicted and observed evolution of TFP following a change in board talent

Notes: The figure plots the evolution of observed and predicted TFP from year-to-event -3 to year-to-event 2 on a balanced subset of firms that (i) change at least one director in year-to-event 0 and (ii) remain in the same tercile of board talent both in the 3 years before the event and in the following 3 years.

Figure A.5: Symmetry of gains and losses in TFP following a change in directors



Notes: The figure plots the change in TFP between the years preeceding and following the event.



Notes: Data are drawn from the combined *Infocamere-Cerved* sample. Joint distribution of (deciles of) firms' TFP and board talent.

Figure A.7: Questionnaire

Invind survey- 2010 wave

For each of the following personnel management practices, specify the extent of their use in your company:				
 use of forms of management remuneration based on results 	none (1); poor (2); moderate (3); high (4)			
• involvement in the decision-making of the lower hierarchical levels	none (1); poor (2); moderate (3); high (4)			

Invind survey- 2019 wave

М	Monitoring section:				
•	What best describes what happens at your firm when a problem in the production process arises? (e.g., finding a quality defect in a service or a product; a piece of equipment breaking down)	we fixed it but did not take further action (1/3); we fixed it and took action to make sure that it does not happen again (2/3); we fixed it and took action to make sure that it does not happen again, and had a continuous improvement process to anticipate problems like these in advance (1); no action was taken (0)			
•	How many key performance indicators are monitored in your firm? (e.g., metrics on production, cost, waste, absenteeism, and quality of services)	1-2 key performance indicators (1/3); 3-9 key performance indicators (2/3); 10 or more key performance indicators; no key performance indicators (0)			
•	How frequently are key performance indicators typically reviewed/updated?	yearly (1/6); quarterly (1/3); monthly (1/2); weekly (2/3); daily (5/6); hourly or more frequently (1); never (0)			
Та	Targets section:				
•	What best describes the time frame of operational targets at your firm?	short-term (less than one year) targets (1/3); long-term (more than one year) targets (2/3); combination of short-term and long-term targets (1); no targets (0)			
•	How easy or difficult typically is it for people to achieve their operational targets in your firm?	without much effort (0); with some effort $(1/2)$; with normal amount of effort $(3/4)$; with more than normal effort (1) ; with extraordinary effort $(1/4)$			
Incentives section:					
•	What are performance bonuses usually based on in your firm?	own performance (1); team performance (3/4); local establishment's or branch's performance (1/2); entire company's performance (1/4); no performance bonuses (0)			
•	What is the primary way workers are promoted in your firm?	solely on performance and ability (1); partly on performance and ability, and partly on other factors (e.g., tenure or family connections) (2/3); mainly on factors other than performance and ability (e.g., tenure or family connections) (1/3); normally no promotions (0)			
•	When is an under-performing worker usually reassigned or dismissed?	within 6 months of identifying worker under-performance (1); after 6 months of identifying worker under-performance (1/2); rarely or never (0)			

The score associated to each answer is reported in parenthesis.